Author's response to reviews

Title: Updating systematic reviews - what difference does it make? Case study of nicotine replacement therapy

Authors:

- Lindsay F Stead (lindsay.stead@dphpc.ox.ac.uk)
- Tim Lancaster (tim.lancaster@dphpc.ox.ac.uk)
- Christopher A Silagy (chris.silagy@med.monash.edu.au)

Version: 2 Date: 5 Apr 2001

Final sentence of background section now reads:
To determine whether the relative effects of treatment were affected by differences in case mix, level of support or length of follow-up.
('level of support' had been omitted)